Author's response to reviews

Title: Primary extra-gastrointestinal stromal tumor of the whole abdominal cavity, omentum, peritoneum and mesentery: a rare case report and review of the articles.

Authors:

Abdulmaged M Musaad (magid999@live.com)
Elsaggad E A (sagadgady@yahoo.com)
Abdulmunem A Abdo (munem2002@hotmail.com)
Ahmed M Elhassan (ahmedelhassan85@gmail.com)
Hiba Hassan (hibahassan@hotmail.com)
Nasreeldeen Adam (sulieman.nasreeldeen@facebook.com)
Mohamed Abdelazeem (mohamedazeem3@hotmail.com)
Nassir A Arabi (nassir_alhaboob@yahoo.com)

Version: 5 Date: 27 April 2014

Author's response to reviews:

dear reviewer
in response to your advicement, the following has been done:
english language has been revised
information about endoscopy has been added
information about histopathology has been added
the topic discussing the main issue
general information has been revised
regards...